Title: A retrospective analysis of adrenal crisis in steroid-dependent patients: causes, frequency and outcomes

Authors:
Katherine Gwenda White (katherine.white@uni-wuerzburg.de;katherinegwhite@ntlworld.com)

Version: 3 Date: 12 Nov 2019

Author’s response to reviews:

1. In your 'Ethics Approval and Consent to Participate' section, you state that consent or ethics approval was not required under the regulatory provisions of the supervisory institutions - please clearly state these regulations and the full names of each supervisory institution.

Done. I’ve provided links rather than quoting section numbers for the 3 ethics institutions we checked against; note that the German regulations come in several parts and that the UK Medical Research Council requires you to self-check online before approaching them. Because this research project was not funded by any institution but conducted entirely with volunteer labour, endorsed by an umbrella coalition of charities, the best advice we could get was that it should be informally supervised by a panel of endocrine specialists.

2. Please ensure that the abstract contains only the following headings: Background, Methods, Results and Conclusions.

Done

3. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Done